Author's response to reviews

Title: Risk factors of acute and overuse musculoskeletal injuries among young conscripts: a population-based cohort study.

Authors:

Henri Taanila (taanila.henri.p@student.uta.fi)
Jaana H Suni (jaana.h.suni@uta.fi)
Pekka Kannus (pekka.kannus@uta.fi)
Harri Pihlajamäki (harri.pihlajamaki@uta.fi)
Juha-Petri Ruohola (juha-petri.ruohola@mil.fi)
Jarmo Viskari (jarmo.viskari@mil.fi)
Jari Parkkari (jari.parkkari@uta.fi)

Version: 6  Date: 1 April 2015

Author's response to reviews: see over
Dear Editor,

Thank you for your e-mail including the comments on our manuscript entitled “Risk factors of acute and overuse musculoskeletal injuries among young conscripts: a population-based cohort study”.

We have revised the manuscript according to the Senior Editor’s suggestions. Our point-by-point responses are given below.

We hope that the revision and responses given are satisfactory and that the revised manuscript is now acceptable for publication in the BMC Musculoskeletal Disorders.

We appreciate the opportunity to revise the manuscript and look forward to hearing from you soon.

Yours sincerely,

Henri Taanila

1) Please clarify whether the obtained consent was verbal or written, kindly indicate this on the Method section of your manuscript
- *This has now been added in Methods section, page 5; line 2.*

2) Indicate in the manuscript whether specific informed consent to publish was obtained from the individual in the figures
- *Specific informed consent was obtained from the person to publish the figures. This has now been added in Methods section page 5, second paragraph, line 20.*

3) Include table 1 in the manuscript as this is absent
- *Thank You for noticing the missing Table 1. This has now been added as presented in previous versions of the manuscript.*